%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Pin No.: 06293
% Paper ID: CDC02-INV2302
% Title: Warga derivate containers and other generalized differentials
% Paper Accepted Category: REGULAR INVITED
% Session Title: Geometric Methods in Control Theory, II: Special Sessions
% Dedicated to Jack Warga's 80th Birthday
% Session Code: FrP11-2
% Presentation Format: ORAL
%
% The full Author's Kit is available for you at the CDC02 web site:
% http://www.eecs.tulane.edu/cdc02/
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  
%            This is a LateX file, containing  the paper 
%
%   WARGA DERIVATE CONTAINERS AND OTHER GENERALIZED DIFFERENTIALS
%
%                       by H. J. Sussmann
%
%        This paper has been submitted for presentation at the 41th IEEE 
%        Conference on Decision and Control, to be held in Las Vegas,
%        December, 2002,  and  for publication in the conference  proceedings. 
%  
%               Date of this version: February 21, 2002
%
%            THIS VERSION SUPERSEDES ALL PREVIOUS VERSIONS
%
%                  The paper is 5 printed pages long.%
%       You should run Latex twice on this file. After the second run,
%      there will still be several ``underfull  hbox'' messages.  Just
%             ignore them and print the .dvi file.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass{article}
\newcounter{pagono}
\setcounter{pagono}{0}
\def\veceps{\vec{\eps}} 
\def\tmath{\mathbb{T}}
\def\nninf{\nn\cup\{+\infty\}}
\def\aof#1{\mathcal{A}(#1)}
\def\llist#1{
\vspace{-4.5mm}
\begin{center}
$\quad\quad\,\,$\mbox{\begin{minipage}{12.2cm}
#1u
\end{minipage}}\end{center}}
\def\pigd{\mbox{path}-\mbox{integral} \mbox{generalized} \mbox{differential}}
\def\pigds{path-integral generalized differentials}
\def\Pigd{\mbox{Path}-\mbox{integral} \mbox{generalized}
\mbox{differential}}
\def\Pigds{\mbox{Path}-\mbox{integral} \mbox{generalized}
\mbox{differentials}}
\def\pigdab{PIGD}
\def\applim{{\rm app\!\!-\!\!lim}}
\def\neigh{\mathcal{Q}_2(\delta)}
\def\neigho{\mathcal{Q}_1(\delta)}
\def\msr{{\rm meas}}
\def\carby{\subset\kern-3pt\subset}
\def\Om{\Omega}
\def\ffi{\varphi}
\def\mybox{\proofend}
\def\traj{{\rm Traj\,}}
\def\ctraj{{\rm CTraj\,}}
\def\arc{{\rm Arc\,}}
\def\carc{{\rm CArc\,}}
\def\bloi{\mathcal{BL}(\Omega,I)}
\def\blxr{\mathcal{BL}(X,\rr)}
\def\blexr{\mathcal{BL}_e(X,\rr)}
\def\bls{\mathcal{BL}(S)}
\def\bles{\mathcal{BL}_e(S)}
\def\mbloiy{\mathcal{M}_e(\bloi,Y)}
\def\meblxry{\mathcal{M}_{\mathcal{BL}}^e(X,\rr;Y)}
\def\mblxry{\mathcal{M}_{\mathcal{BL}}(X,\rr;Y)}
\def\mloceblxry{\mathcal{M}_{\mathcal{BL}}^{e,loc}(X,\rr;Y)}
\def\bbxr{\mathcal{B}(X\times\rr)}
\def\cvf{TVVF}
\def\cvfadomi{\mathcal{ADM}}
\def\comment#1{}
%\def\brokarrow{\,- \to}  
\def\brokarrow{\,\hbox to6mm{-\hfill-\hfill-\hfill-\hfill}\!\!\!\!>\,}
\def\traj{{\rm Traj\,}}
\def\cvfadomi{\mathcal{ADM}}
\def\comp{\mathbb{C}}
\def\disc{\mathbb{D}}
\def\tub{\mathcal{T}}
\def\tubx{\mathcal{T}^X}
\def\tubn{\mathcal{T}^n}
\def\tubxp{\mathcal{T}^{X,+}}
\def\tubxc{\mathcal{T}^{X,C}}
\def\tubxm{\mathcal{T}^{X,-}}
\def\tubxpm{\mathcal{T}^{X,\pm}}
\def\btub{\bar\mathcal{T}}
\def\btubx{\bar\mathcal{T}^X}
\def\linx{{\rm Lin}(X;X)}
\def\linxxp{{\rm Lin}(X;X')}
\def\linxy{{\rm Lin}(X;Y)}
\def\Applim{{\rm App\!\!-\!\!lim}}
\def\applim{{\rm app\!\!-\!\!lim}}
\def\cvf{TVVF}
\def\id{\kern.2em{\rm I}\kern-.56em{\rm 1}}
\def\restr{\,\lceil\,}
\def\czer#1#2{C^0(\,#1\,;\,#2\,)}
\newcommand{\nball}{{\bar\bmath}^n}
\def\traj{{\rm Traj\,}}
\def\ctraj{{\rm CTraj\,}}
\def\arc{{\rm Arc\,}}
\def\carc{{\rm CArc\,}}
\def\carby{\subset\kern-3pt\subset}
\def\ffi{\varphi}
\def\brokarrow{\,\hbox to6mm{-\hfill-\hfill-\hfill-\hfill}\!\!\!\!>\,}
\def\nexr{\mathcal{N}(X,\rr)}
\def\nrnr{\mathcal{N}(\rr^n,\rr)}
\def\bloi{\mathcal{BL}(\Omega,I)}
\def\blxr{\mathcal{BL}(X,\rr)}
\def\blexr{\mathcal{BL}_e(X,\rr)}
\def\blernr{\mathcal{BL}_e(\rr^n,\rr)}
\def\blrnr{\mathcal{BL}(\rr^n,\rr)}
\def\blrn{\mathcal{B}(\rr^n)}
\def\bls{\mathcal{BL}(S)}
\def\bles{\mathcal{BL}_e(S)}
\def\lmblernrm{\mathcal{M}_{\mathcal{BL}}^{e,loc}(\rr^n\times\rr;\rr^m)}
\def\mbloiy{\mathcal{M}_e(\bloi,Y)}
\def\meblxry{\mathcal{M}_{\mathcal{BL}}^e(X,\rr;Y)}
\def\mblxry{\mathcal{M}_{\mathcal{BL}}(X,\rr;Y)}
\def\mloceblxry{\mathcal{M}_{\mathcal{BL}}^{e,loc}(X,\rr;Y)}
\def\bbxr{\mathcal{B}(X\times\rr)}
\def\so{{\rm So}} 
\def\ta{{\rm Ta}}
\def\gr{{\rm Gr}}
\def\dom{{\rm Do}}
\def\im{{\rm Im}}
\def\bem{\em }
\def\togr{\,\,{\buildrel \rm igr \over \longrightarrow}\,\,}
\def\tto{\mbox{$\rightarrow\kern-6pt\rightarrow\,$}}
\def\regu#1#2{{\rm REG}(#1;#2)}
\newcommand{\skipopaghard}{\stepcounter{pagono}\newpage \null

\vspace{3cm}

{\Large  

\noindent This is the back of {\Huge Page \arabic{pagono}}\\
of the paper

\vspace{1cm}
\begin{center}
{\huge Warga derivate containers\\ 
and other generalized\\
differentials}

$\quad$\\

by\\

$\quad$\\ 
{\huge H\'ector J. Sussmann}\\

\end{center}
}

\newpage 

\null

\vspace{3.3cm}

\noindent Paper ID: $\,\,\,${\huge  CDC02-INV2302} \\

\noindent Session Title:\\

\medbreak
{\Large \noindent Geometric Methods in Control\\

\vspace{-5mm}\noindent Theory, II: Special Sessions\\

\vspace{-5mm}\noindent Dedicated to Jack Warga's\\

\vspace{-3mm}\noindent 80th Birthday}

\vspace{4mm}
Session Code: {\Huge FrP11-2} \\

Paper Category: REGULAR INVITED

\newpage
}
\usepackage{amsfonts}
\newcommand{\parnew}{\parindent=1cm}
\newcommand{\vup}{\vspace{-1mm}}
\newcommand{\vupp}{\vspace{-2mm}}
\newcommand{\vuppp}{\vspace{-3mm}}
\newcommand{\vupppp}{\vspace{-4mm}}
\newcommand{\vuppppp}{\vspace{-5mm}}
\def\equaldef{{\buildrel \rm def \over =}}
\def\proofend{\hfill$\diamondsuit\,\,$}
\def\pmb#1{\setbox0=\hbox{$#1$}%
  \kern-.025em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.025em\raise.0433em\box0 }
\def\pmbi#1{\setbox0=\hbox{$#1$}%
  \kern-.01em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.01em\raise.013em\box0 }
\def\pmbo#1{\setbox0=\hbox{$#1$}%
  \kern-.055em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.055em\raise.0433em\box0 }
\def\zz{\mathbb{Z}}
\def\qq{\mathbb{Q}}
\def\cc{\mathbb{C}}
\def\rr{\mathbb{R}}
\def\nn{\mathbb{N}}
\def\kk{\mathbb{K}}
\def\bmath{\mathbb{B}}
%AFTER begin doc, redefined abovedisplayskip, etc.

\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\C}{{\cal {C}}}
\newcommand{\E}{{\cal {E}}}
\renewcommand{\L}{{\cal {L}}}
\newcommand{\Q}{{\cal {Q}}}
\newcommand{\M}{{\cal {M}}}
\newcommand{\F}{{\cal {F}}}
\newcommand{\U}{{\cal {U}}}
\newcommand{\T}{{\cal {T}}}
\newcommand{\J}{{\cal {J}}}
\newcommand{\bed}{\begin{description}}
\newcommand{\eed}{\end{description}}
\newtheorem{Theorem}{Theorem}
\newtheorem{Principle}{Principle}
\newtheorem{Proposition}{Proposition}
\newtheorem{Lemma}{Lemma}
\newtheorem{Definition}{Definition}
\newtheorem{Remark}{Remark}
\newtheorem{Example}{Example}
\def\deq{\stackrel{def}{=}}
%\def\restriction{\uprightharpoon}
\def\restriction{\uparrow}
\def\n{\noindent}
\def\vs{\vskip 1em}
\def\R{\rr}
\def\N{\nn}
\def\Z{\zz}
\def\ve{\varepsilon}
\def\al{\alpha}
\def\vp{\varphi}

\input amssym.def

\newcommand{\notinproceedings}[1]{}  %material to omit in short version
%\newcommand{\notinproceedings}[1]{#1}  %material to include in long version
\newcommand{\inproceedings}[1]{#1}  %material to include in short version
%\newcommand{\inproceedings}[1]{}  %material NOT to include in long version

\newcommand{\tempnote}[1]{{\large \bf #1}}
\newcommand{\good}{DBCBP}
%\newcommand{\tempnote}[1]{}

%\newcommand{\pubnote}{\vskip-5.0cm
%\framebox{Proc.\ {\it 95 CDC- Keywords: stabilization, nonlinear, Lyapunov functions}} 
%\vskip4.5cm}

\renewcommand{\baselinestretch}{0.84} %to fine-tune - use .95 if you need more
\vspace{-15pt}
\def\email{}    % in case you don't want to do this
\newcommand{\autwo}{% Fill this in for a three-author paper
		       % Of course, footnote marks may differ
\begin{tabular}[t] {c@{\extracolsep{0.5em}}c@{\extracolsep{0.5em}}c}
\vspace{-15pt}
\begin{minipage}[t]{4.9in}
\begin{center}
{\Large\em Dedicated to Jack Warga on his 80th birthday}\\
$\,$\\
H\'ector J.\ Sussmann\\
Department of Mathematics\\
Rutgers, the State University of New Jersey\\
Hill Center---Busch Campus\\
%110 Frelinghuysen Road\\
Piscataway, NJ 08854-8019, USA\\
{\small\tt sussmann@math.rutgers.edu}\\
$\qquad$\\

\end{center}
\end{minipage}
\end{tabular}
\vskip5pt
} %end of definition of \autwo

%AUTHORS:  (UNCOMMENT JUST ONE OF THE NEXT THREE LINES)
\newcommand{\auone}{
\vspace{-15pt}
\begin{center}
$\,$\\
H\'ector J. Sussmann\\
%\footnote[1]{\supportnote}\\
%\footnote{\supportnote}\\
Department of Mathematics; Rutgers, the State University of New Jersey\\
Hill Center---Busch Campus, 110 Frelinghuysen Road, Piscataway, NJ
08854-8019, USA\\ 
\vskip5pt
E-mail: sussmann@hamilton.rutgers.edu\quad\quad
Web site: http://www.math.rutgers.edu/$\widetilde{\,\,}$sussmann\\
\end{center}
\vskip5pt
} 


\newcommand{\dato}{February 26, 2001}

\newcommand{\authors}{
%\auone
\autwo
%\authree
}
\renewcommand{\thefootnote}{\fnsymbol{footnote}}
\renewcommand{\title}{
\vspace{-1cm}
{\bf Warga derivate containers and other generalized differentials$\,^{*}$}
}

%% THE FOLLOWING SPACING COMMANDS MAY HAVE TO BE FINE-TUNED DEPENDING ON
%% ADDRESSES LENGTH:
\def\spaceabovetitle{\hspace{1cm}}
\def\spacebelowtitle{\vspace{10pt}}
\def\spaceafterauthors{}
%\def\spaceafterauthors{\vspace{-1cm}}


\newcommand{\topofpageonenosubmission}{
%\hfill {\Large Paper CDC00-INV3301}
}

\newcommand{\topofpageoneforsubmission}{{\small \null
\vspace{-9mm} $\quad$ \\ $\quad$}}

\def\titlestuff{\topofpageonenosubmission

\vspace{-5.6mm}
\null\vspace{4mm}
\vspace{2mm}
\begin{center}
\spaceabovetitle
\begin{center}
{\Large \bf \title}\\
\end{center}
\spacebelowtitle
\authors
\spaceafterauthors
\end{center}}

\tolerance=20 %allow VERY LITTLE tolerance in extending after line.
\overfullrule=0pt %no dark lines if overfull

%CHANGED BY EDS, July 95 - had problems with xref before
\newcommand{\ssection}[1]{\vspace{0.15cm}%
   \refstepcounter{section}%
   \centerline{{\bf \thesection.~#1}}\vspace{0.2cm}}
\newcommand{\subssection}[1]{\vspace{0.15cm}%
   \refstepcounter{subsection}%
  \noindent{{\bf\thesubsection.~#1}}\vspace{0.1cm}} 
%\newcommand{\ssection}[1]{\vspace{0.15cm}\setcounter{subsection}{0}
%  \protect\addtocounter{section}{1}%
%  \centerline{{\bf \thesection.~#1}}\vspace{0.2cm}}
%\newcommand{\subssection}[1]{\vspace{0.15cm}
%  \protect\addtocounter{subsection}{1}
%  \noindent{{\bf\thesubsection.~#1}}\vspace{0.1cm}} 

\newcommand{\skipopag}{\newpage}
%\topmargin -2.3cm
\topmargin -1.7cm%COMMENT THIS OUT FOR HOME PRINTING
%\topmargin -0.5cm%USE ONLY FOR HOME PRINTING
%COMMENT THIS OUT FOR REGULAR PRINTING BUT USE IT FOR HARDCOPY SUBMISSION
%\pagestyle{empty}\renewcommand{\skipopag}{\skipopaghard}
%COMMENT THIS OUT FOR REGULAR PRINTING BUT USE IT FOR ELECTRONIC SUBMISSION
%\pagestyle{empty}
\textheight 23.9cm
\oddsidemargin -0.3in
\evensidemargin -0.3in
\textwidth 177mm

\sloppy
\flushbottom
\parindent 1em
\leftmargini 2em
\leftmarginv .5em
\leftmarginvi .5em

\marginparwidth 48pt 
\marginparsep 10pt 
\columnsep 7mm

\newcommand{\halmos}{\rule{1ex}{1.4ex}}
%\newenvironment{proof}{\noindent {\em Proof}.\ }
%{\hspace*{\fill}$\halmos$\medskip}
\newenvironment{proof}{\noindent {\em Proof}.\ }
{\proofend\medskip}
\newtheorem{itlemma}{Lemma}[section] %number by section (set in \em by default)
%same numbering counter as itlemmas:
\newcommand{\compo}{\circ}
\newtheorem{theorem}[itlemma]{Theorem}
\newtheorem{itproposition}[itlemma]{Proposition}
\newtheorem{itcorollary}[itlemma]{Corollary}
\newtheorem{itremark}[itlemma]{Remark}
\newtheorem{itdefinition}[itlemma]{Definition}
\newtheorem{itexample}[itlemma]{Example}
\newenvironment{lemma}{\begin{itlemma}}{\end{itlemma}} %no-italics
\newenvironment{remark}{\begin{itremark}\rm}{\end{itremark}} %no-italics
\newenvironment{corollary}{\begin{itcorollary}}{\end{itcorollary}}
\newenvironment{proposition}{\begin{itproposition}}{\end{itproposition}}
\newenvironment{definition}{\begin{itdefinition}\rm}{\end{itdefinition}}
\newenvironment{example}{\begin{itexample}\rm}{\end{itexample}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% These are for choices.  They take 4 or 6 args
%% (carriage returns allowed in between}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\twoif}[4]{
\left\{ \begin{array}{ll}#1&#2\\#3&#4\end{array}\right.
}
\newcommand{\threeif}[6]{
\left\{ \begin{array}{ll}#1&#2\\#3&#4\\#5&#6\end{array}\right.
}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\beqn}{\begin{eqnarray}}
\newcommand{\eeqn}{\end{eqnarray}}

\newcommand{\rf}[1]{[\ref{#1}]} %bibliographical references

\newcommand{\text}[1]{\hbox{\rm \ #1\ \/}}
\newcommand{\be}[1]{\begin{equation}\label{#1}}
\newcommand{\ee}{\end{equation}}
\newcommand{\bl}[1]{\begin{lemma}\label{#1}}
\newcommand{\br}[1]{\begin{remark}\label{#1}}
\newcommand{\bt}[1]{\begin{theorem}\label{#1}}
\newcommand{\bd}[1]{\begin{definition}\label{#1}}
\newcommand{\bp}[1]{\begin{proposition}\label{#1}}
\newcommand{\bc}[1]{\begin{corollary}\label{#1}}
\newcommand{\ber}[1]{\begin{exercise}\label{#1}}
\newcommand{\bex}[1]{\begin{example}\label{#1}}
%\newcommand{\bem}[1]{\begin{example}\label{#1}}  %Yes, 2 different ones...
\newcommand{\ec}{\mybox\end{corollary}}
\newcommand{\el}{\mybox\end{lemma}}
\newcommand{\er}{\mybox\end{remark}}
\newcommand{\et}{\qed\end{theorem}}
\newcommand{\ed}{\mybox\end{definition}}
\newcommand{\ep}{\mybox\end{proposition}}
\newcommand{\epr}{\end{proof}}
\newcommand{\bpr}{\begin{proof}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  These are just for statements, no "qed" box is put in.
%%  (Also for when I have to finish something after a displayed equation
%%        or itemized.)
\newcommand{\ecs}{\end{corollary}}
\newcommand{\eers}{\end{exercise}}
\newcommand{\eexs}{\end{example}}
\newcommand{\eems}{\end{example}}
\newcommand{\els}{\end{lemma}}
\newcommand{\eles}{\end{lemmaex}}
\newcommand{\ers}{\end{remark}}
\newcommand{\ets}{\end{theorem}}
\newcommand{\eds}{\end{definition}}
\newcommand{\eps}{\varepsilon}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\qed}{\hfill \mbox{$\halmos$}} %put qed at right margin

\newcommand{\X}{\Bbb X} % state space
\newcommand{\Prob}{\Bbb P} % Probability measures
\newcommand{\Rp}{{\Bbb R}_{\geq 0}} % nonnegative reals

\newcommand{\zp}{0+}

\newcommand{\dd}[1]{D_{#1}^-}

\newcommand{\inter}{{\rm int \,}}
\newcommand{\rank}{{\rm rank \,}}
%\newcommand{\im}{{\rm im \,}}
%\newcommand{\ker}{{\rm ker \,}}
\newcommand{\norm}[1]{\left\Vert #1 \right\Vert}
\newcommand{\normu}[1]{\norm{#1}} %in case I decide to change to sub-infty
\newcommand{\Gc}{{{\cal S}}}
\newcommand{\hatV}{\widehat{V}}

%\newcommand{\coco}[1]{{\langle #1 \rangle}} %closed convex hull
\newcommand{\coco}[1]{{\overline{{\rm co}}( #1 )}} %closed convex hull

\newcommand{\abs}[1]{\left\vert #1 \right\vert}
\def\st{\, | \,}
\newcommand{\hsn}{\hspace{-1.5pt}}


%%%%%%%%%%%%%%% macros for this paper %%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\mylimsup}[1]{\raisebox{-1ex}{$\stackrel{\overline{\displaystyle\lim}}{\scriptscriptstyle #1}$}}

%\newcommand{\myliminf}[2]{\liminf_{\stackrel{{\scriptstyle #1}}{#2}}}
\newcommand{\myliminf}[2]{\liminf_{\stackrel{{\textstyle #1}}{\textstyle #2}}}

\newcommand{\opteq}[1]{\[#1\]}
\newcommand{\rref}[1]{(\ref{#1})}
\newcommand{\any}{\forall}
\newcommand{\clf}{CLF}

\newcommand{\bi}{\begin{itemize}}
\newcommand{\ei}{\end{itemize}}
\newcommand{\ben}{\begin{enumerate}}
\newcommand{\een}{\end{enumerate}}

\begin{document}
\begin{thebibliography}{A}

\vspace{-3mm}
\begin{thebibliography}{99}

\bibitem{berkovitz}{Berkovitz,~L.~D., \textit{Optimal Control Theory.}
Springer-Verlag, New York, 1974.}

\bibitem{botsko}{Botsko,~M.~W., and R.~A.~Gosser, \textit{On the 
differentiability of functions of several variables.} Amer. Math. 
Monthly \textbf{92}, No. 9, 1985, pp.~663--665.}

\bibitem{halkina}{Halkin,~H., \textit{Necessary conditions in
mathematical programming and optimal control theory.} In
\textit{Optimal control theory and its applications, Part I,}
\underline{Lect. Notes in Econom. and} \underline{Math. Systems}
\textbf{105}, Springer-Verlag, Berlin, 1974, pp.~113--165.}

\bibitem{halkinb}{Halkin,~H., \textit{Interior mapping theorem with
set-valued derivatives.} Journal d'Analyse Math\'emathique
\textbf{30}, 1976, pp.~200--207.}

\bibitem{halkinc}{Halkin,~H., \textit{Necessary conditions for optimal
control problems with differentiable or nondifferentiable data.} In
\textit{Mathematical Control Theory}, \underline{Lect. Notes in Math.}
\textbf{680}, Springer-Verlag, Berlin, 1978, pp.~77--118.}

\bibitem{pontrybook}{Pontryagin,~L.~S.,~V.~G.~Boltyanskii,~R.~V.~Gamkrelidze,
and E.~F.~Mischenko, \textit{The Mathematical Theory of Optimal Processes.}
Wiley, New York, 1962.}

\bibitem{susspavel} {Sussmann,~H.~J., \textit{A strong version of the
Lojasiewicz maximum principle.} In \textit{Optimal Control of Differential 
Equations,} N.~H.~Pavel Ed., Marcel Dekker Inc., New York, 1994,
pp.~293--309.}

\bibitem{susskobe} {Sussmann,~H.~J.,  
\textit{A strong maximum principle for systems of differential inclusions.}
In \textit{Proc. 35th IEEE Conference on Decision and Control,
Kobe, Japan, Dec. 1996,} IEEE publications, 1996,  pp.~1809--1814.}

\bibitem{sussrecent} {Sussmann,~H.~J.,  \textit{Some recent results on 
the maximum principle of optimal control theory.}
In \textit{Systems and Control in the Twenty-First Century,} 
C. I. Byrnes, B. N. Datta, D. S. Gilliam and C. F. Martin Eds.,
Birkh\"auser, Boston, 1997, pp.~351--372.}

\bibitem{sussmul}{Sussmann,~H.~J., \textit{Multidifferential calculus: chain
rule, open mapping and transversal intersection theorems.} In \textit{Optimal
Control: Theory, Algorithms, and Applications,} W. W.  Hager and
P. M. Pardalos, Editors, Kluwer Academic Publishers, 1998, pp.~436--487.}

\bibitem{sussshyb}{Sussmann,~H.J., ``A maximum principle
for \mbox{hybrid} optimal control problems.'' In \textit{Proc. 38th IEEE
Conf. Decision and Control, Phoenix, AZ, Dec. 1999}.  IEEE
publications, New York, 1999, \mbox{pp.~425--430}.} 

\bibitem{susssmf2000}{Sussmann,~H.J., ``R\'esultats r\'ecents
sur les courbes optimales.''  In \textit{15$\,^e$ Journ\'ee Annuelle
de la Soci\'et\'e Math\'emathique de France (SMF)}, Publications de
la SMF, Paris, 2000, pp.~1-52.}

\bibitem{sussncn2000}{Sussmann,~H.J., ``New theories of set-valued
\mbox{differentials} 
and new versions of the maximum \mbox{principle} of optimal control 
theory.'' In \textit{Nonlinear Control in the year 2000},
A.~Isidori, F.~Lamnabhi-Lagarrigue and  W.~Respondek Eds.,
Springer-Verlag, London, 2000, pp.~487-526.}

\bibitem{susssyd2000}{Sussmann,~H.J., 
``Set-valued differentials and the hybrid maximum principle.''  
In \textit{Proc. 39th IEEE Conf. Decision
and Control, Sydney, \mbox{Australia}, \mbox{December} 12-15, 2000}, IEEE
\mbox{publications}, \mbox{New York}, 2000.}

\bibitem{jct}{Tougeron, J.-C., \textit{Id\'eaux de fonctions 
diff\'erentiables}. Ergebnisse der Mathematik und ihrer
Grenzgebiete, Band 71. Springer-Verlag, 1972.}

\bibitem{wargaa}{Warga,~J., \textit{Fat homeomorphisms and unbounded derivate containers}.  J. Math. Anal. Appl. \textbf{81}, 1981, pp.~545--560.}

\bibitem{wargab}{Warga,~J., \textit{Controllability, extremality
and abnormality in nonsmooth optimal control.} J. Optim. Theory
Applic. \textbf{41}, 1983, pp.~239--260.}

\bibitem{wargac}{Warga,~J., \textit{Optimization and controllability without differentiability  assumptions.}  SIAM J. Control and Optimization \textbf{21}, 1983, pp.~837--855.}

\bibitem{wargad}{Warga,~J., \textit{Homeomorphisms and local $C^1$ approximations}. Nonlinear Anal. TMA \textbf{12}, 1988, pp.~593--597.}

\bibitem{wargae}{Warga,~J., \textit{An extension of the Kaskosz maximum principle}. Applied Math. Optim. \textbf{22}, 1990, pp.~61--74.}
 
\end{thebibliography}


\abovedisplayskip 4pt plus2pt minus5pt\belowdisplayskip \abovedisplayskip
\twocolumn[\titlestuff]
\footnotetext[1]{The  author was supported in part by
NSF Grant DMS98-03411-00798.}

\vskip4.5cm

\ssection{Introduction}
\label{sec-intro}

\noindent This is the first of two papers devoted to recent ideas
on the theory of generalized differentials with good open mapping properties.
Here we will discuss ``generalized differentiation 
theories'' (abbr. GDTs), 
with special 
\mbox{emphasis} on the series of developments initiated 
by Jack Warga's pioneering work on derivate containers.
In the second paper, we will focus on the most recent theory,
of ``path-integral generalized differentials,'' and prove that
it has the \mbox{crucial} properties required for a version of the 
Pontryagin \mbox{Maximum} \mbox{Principle} (\mbox{abbr.} PMP) 
to exist, namely, the chain rule and the directional open
\mbox{mapping} property. 

Our work continues the \mbox{study}
of general smooth, \mbox{nonsmooth}, high-order, and \mbox{hybrid} 
\mbox{versions} of the \mbox{PMP} for finite-dimensional
\mbox{deterministic}
\mbox{optimal} \mbox{control} \mbox{problems} without state \mbox{space}
\mbox{constraints}
by \mbox{means} of a method \mbox{developed} by us in recent years.
As \mbox{explained} in
\cite{sussshyb,susssmf2000,sussncn2000,susssyd2000}, 
such \mbox{versions} can be \mbox{derived} in a \mbox{unified} way, by using
a \mbox{modified} version of the \mbox{approach} of the classical book
\cite{pontrybook} by 
\mbox{Pontryagin} {\it et al.}. In the classical approach,
one
\mbox{constructs} ``\mbox{packets} of needle \mbox{variations},'' 
\mbox{linearly} \mbox{approximates} these
\mbox{packets} at the base \mbox{value} of the \mbox{variation} parameter,
and \mbox{propagates} the \mbox{resulting} \mbox{linear} approximations to the
terminal point of the \mbox{trajectory} by means of the \mbox{differentials}
of the \mbox{reference} flow maps.  This technique
must be \mbox{modified} by \mbox{(a)~replacing}
the the theory of the \mbox{classical} 
\mbox{differential} by
other GDTs, \mbox{(b)~replacing} the time-varying vector fields that
\mbox{occur} in the classical MP by {\em flows}, and (c)~replacing
the \mbox{needle} variations by {\em \mbox{abstract}
variations}.  Every GDT yields a \mbox{version} of the MP
provided it \mbox{satisfies}
some \mbox{natural} properties such as the chain rule
and an \mbox{appropriate} ``\mbox{directional} open 
\mbox{mapping} property'' (abbr. DOMP).
(\mbox{Details} are provided in
\cite{sussshyb,susssmf2000,sussncn2000,susssyd2000}.)

In these two papers we will not explicitly
discuss flows, variations, and the
PMP. How these topics are dealt with in the theory, and how
one gets versions of the PMP for general GDTs, has been extensively
disucssed in other papers, and for lack of space we choose not to address
these questions here. We will, instead, give a detailed
discussion of GDTs, and show how several important GDTs
are defined and how they are related to one \mbox{another}.
It will turn out that the complicated nature of these interrelationships,
and in particular the fact that none of the ``natural'' theories
contains all the others, suggests the question whether
a theory that truly contains all other GDTs can be defined.
The answer to this question is given in the second paper of the series,
where a theory that does the desired job is defined and studied.

Here we propose an axiomatic definition of the \mbox{concept} of a GDT
and a precise statement of the 
\mbox{DOMP}. We then outline the definitions of
some \mbox{GDTs}, such as the Warga derivate containers (abbr. 
\mbox{WDCs}), and our more recent theories of 
\mbox{``multidifferentials''} and  ``\mbox{generalized}
differential quotients'' (abbbr. GDQs).  Multidifferentials are
generalizations of Warga's derivate containers, which in turn include
the Clarke \mbox{generalized} Jacobians (abbr. CGJs) as a special case. The
GDQs, on the other hand, generalize the CGJs in a \mbox{different}
\mbox{direction}, and constitute a 
\mbox{theory} that \mbox{neither} 
\mbox{contains} nor is contained in that
of the WDCs or in that of \mbox{multidifferentials.} We will make this
precise by \mbox{comparing} these two types of theories by means
of \mbox{examples}. This will set the stage for
the second \mbox{paper}, in which we will presents a theory that achieves the
\mbox{desired} unification.

\begin{remark}
Generalized differentials that extend the classical differential
and the Warga derivate containers were studied by H. Halkin
in the 1970s in three remarkably insightful papers 
(\cite{halkina, halkinb, halkinc}) that, unfortunately, 
do not seem to have attracted the attention they deserved.

The work presented here is a continuation and extension of 
that of Halkin. In particular, our machinery makes it possible to deal 
systematically with set-valued maps, which appear when one studies
flows of continuous but not necessarily Lipschitz vector fields
(as in the ``\L ojasiewicz maximum principle,'' cf. 
\cite{susspavel}) 
and also in the analysis of differential inclusions (cf.
\cite{susskobe}). \proofend
\end{remark}





\ssection{GDTs}\label{gdts}

\noindent A ``generalized differentiation theory'' (abbr. GDT) is, roughly,
a way of assigning a ``differential'' $\mathcal{D}(F;x,y;S)$ 
to each 4-tuple $(F,x,y,S)$ consisiting
of (a)~a set valued map $F:M\tto N$ whose source $M$ and target 
$N$ are manifolds of class $C^1$,
(b)~a point $x$ of $M$, (c)~a point $y$ of $N$, and (d)~a subset
$S$ of $M$.  The object $\mathcal{D}(F;x,y;S)$ is required
to be a set of nonempty compact subsets of ${\rm Lin}(T_xM,T_yN)$, the
space of linear maps from $T_xM$ to $T_yN$, where, for a manifold
$Q$ and a $q\in Q$, $T_qQ$ denotes
the tangent space to $Q$ at $q$.
The members of $\mathcal{D}(F;x,y;S)$ are the 
``$\mathcal{D}$-differentials of $F$ at $(x,y)$ in the direction
of $S$.''  (The  set  $\mathcal{D}(F;x,y;S)$ could be empty, in which
case we say that $F$ is not 
$\mathcal{}D$-differentiable at $(x,y)$ in the direction
of $S$.)  The correspondence $\mathcal{D}$ is required to satisfy the
chain rule, the Cartesian product rule, locality, and invariance under
$C^1$ diffeomorphisms.  Furthermore, it is required to be an extension
of the classical theory of differentials of maps of class $C^1$.

To make all this precise, we introduce some notations and definitions.

\skipopag

\noindent{\em Set-valued maps.}
By a  {\em set-valued map} (abbr. SVM) we mean a triple 
$F=(A,B,G)$ such that $A$ and $B$ are 
\mbox{sets} and $G$ is a subset of
$A\times B$. The \mbox{sets} $A$, $B$, $G$ are, 
\mbox{respectively}, the {\em
source}, {\em \mbox{target}}, and {\em graph} of $F$, and we
write $A=\so(F)$, 
\mbox{$B=\ta(F)$}, \mbox{$G=\gr(F)$}.  If $x$ is any object, we
write \mbox{$F(x)=\{y:(x,y)\in\gr(F)\}$}.  (Hence $F(x)=\emptyset$ unless
\mbox{$x\in\so(F)$}.)  The \mbox{sets}
\mbox{$\dom(F)=\{x\in\so(F):F(x)\neq\emptyset\}$},
$\im(F)=\bigcup_{x\in\so(F)}F(x)$, are, respectively, the {\bem
domain} and {\em image} of $F$. If $F=(A,B,G)$ is an SVM, we
say that $F$ is an {\em SVM from $A$ to $B$}, and write
$F:A\tto B$.  We use $SVM(A,B)$ to denote the set of all SVMs
from $A$ to $B$.
The expression ``ppd map'' stands for ``\mbox{possibly}
\mbox{partially} defined
(that is, not necessarily everywhere \mbox{defined})
ordinary (that is, single-valued) map,'' and we write
$f:A\brokarrow B$ to indicate that $f$ is a ppd map from $A$ to
$B$.  A {\it time-varying ppd map} from a set $A$ to a set $B$ is a
ppd map from $A\times\rr$ to $B$.

\medbreak\noindent{\em Classes of manifolds.}
We use $\zz_+$ to denote the set of all nonnegative
\mbox{integers}. If $k\in\zz_+$, then $\mathcal{M}^k$,
$\mathcal{SVM(M}^k)$ will denote, \mbox{respectively}, the class of
\mbox{all}
finite-dimensional Hausdorff manifolds of class $C^k$ \mbox{without}
boundary, and the class of all \mbox{SVMs} $F$ such that
$\so(F)\!\in\!\mathcal{M}^k$ and $\ta(F)\!\in\!\mathcal{M}^k$.  If
$k\ge 1$ and \mbox{$x\!\!\in\!\! M\!\!\in\!\!\mathcal{M}^k$}, then
$T_xM$, $T_x^*M$, $TM$, $T^*M$ will denote, respectively, the
\mbox{tangent} and cotangent \mbox{spaces} to $M$ at $x$, and the
tangent and cotangent bundles of $M$. Clearly, $TM$ and $T^*M$ belong
to $\mathcal{SVM(M}^{k-1})$.

\medbreak\noindent{\em The Bouligand tangent cone.}
If $M\in\mathcal{M}^1$, $x\in M$, and $S\subseteq M$, then
$T_x^BS$ will denote the {\em Bouligand \mbox{tangent} cone} to $S$ 
at $x$. By \mbox{definition}, $T_x^BS$  is the set of
\mbox{all} tangent vectors $v\in T_xM$ such that there exist a 
\mbox{sequence}
$\{(x_j,h_j)\}_{j\in\nn}$ of points of $S\times\rr$ having the property that
$h_j>0$ for all $j$, $h_j\downarrow 0$ as $j\to\infty$, and 
$\lim_{j\to\infty}h_j^{-1}(\varphi(x_j)-\varphi(x))
=v\varphi$ for all functions $\varphi\in C^1(M,\rr)$.

\medbreak\noindent{\em Classes of linear spaces, linear 
multimaps.}
We write $\mathcal{RLS}$,  $\mathcal{FDRLS}$, 
to denote, respectively,  the class of all linear spaces over $\rr$, and 
the class of all $X\in\mathcal{RLS}$ that are finite-dimensional.
If $X,Y\in\mathcal{RLS}$,  then 
${\rm Lin}(X,Y)$ denotes the set of all $\rr$-linear maps from
$X$ to $Y$. A \mbox{subset} of ${\rm Lin}(X,Y)$ will be called a {\em 
\mbox{linear} \mbox{multimap}} from $X$ to $Y$. 
We use ${\rm MLin}(X,Y)$, ${\rm MLin}_c(X,Y)$,
to denote, \mbox{orespectively}, 
the set of all linear  multimaps from $X$ to $Y$
and (if $X,Y\in\mathcal{FDRLS}$) the set of \mbox{all} nonempty 
compact linear  multimaps from $X$ to $Y$.

\begin{definition}\label{defpfgdt}
A {\em generalized differentiation theory} (abbr. GDT)
is a correspondence $\mathcal{D}$ that \mbox{assigns} to 
\mbox{every} $F\in\mathcal{SVM}(\mathcal{M}^1)$,
every \mbox{$(x,y)\in\so(F)\times\ta(F)$},
and \mbox{every} subset $S$ of the source $\so(F)$, 
a set \mbox{$\mathcal{D}(F;x,y;S)\subseteq {\rm MLin}_c(T_xM,T_yY)$}
of nonempty 
\mbox{compact} linear multimaps from $T_xM$ to $T_yY$,
in such a way that the following axioms are satisfied:

\ben 
\vspace{-6pt}
\item[1.] If $m,n\in\zz_+$, $f\!:\!\rr^m\!\mapsto\!\rr^n$ is a map 
of class $C^1$, and $S$ is a polyhedral convex cone in $\rr^m$, then
$\{Df(0)\}\!\in\!\mathcal{D}(f;0,0;S)$. 

\comment{
\vspace{-8pt}
\item[2.] ({\bem Monotonicity}) Whenever 
$F_1,F_2\in\mathcal{SVM}(\mathcal{M}^1)$,
$M=\so(F_1)=\so(F_2)$, $N=\ta(F_1)=\ta(F_2)$, $x\in M$, $y\in N$, 
$S\subseteq M$, $\Lambda\in\mathcal{D}(F_1;x,y;S)$, 
$\Lambda'\in{\rm MLin}_c(T_xM,T_yN)$, 
$\gr(F_1)\subseteq\gr(F_2)$ and 
$\Lambda\subseteq\Lambda'$, it follows that
$\Lambda'\in\mathcal{D}(F_2;x,y;S)$.
}

\vspace{-8pt}
\item[2.] {\bem (The chain rule)}  If $F_i\!\in\!\mathcal{SVM}(\mathcal{M}^1)$
and \mbox{$S_i\!\subseteq\! M_i$} for $i\!=\!1,2$,  
$M_1\!=\!\so(F_1)$, \mbox{$M_2\!=\!\ta(F_1)\!=\!\so(F_2)$}, 
\mbox{$M_3\!=\!\ta(F_2)$}, $x_i\!\in\! M_i$ for $i\!=\!1,2,3$,
$F_1(S_1)\!\subseteq\! S_2$, $\Lambda_1(T_{x_1}^BS_1)\!\subseteq\!
T_{x_2}^BS_2$, and
$\Lambda_2\!\in\!\mathcal{D}(F_2;x_{2},x_{3};S_2)$, then 
\mbox{$\Lambda_2\circ\Lambda_1\!\in\!\mathcal{D}(F_2\circ F_1;x_{1},x_3;S_1)$}.

\comment{
if either (a)~$F_1$ is a diffeomorphism of class
$C^1$ from a neighborhood $U_1$ of $x_1$ onto a neighborhood
$U_2$ of $x_2$, $F_1(x_1)=x_2$, $F_1(S_1\cap U_1)=S_2\cap U_2$, 
and \mbox{$\Lambda_1=\{DF_1(x_1)\}$}, or
(b)~$F_1$ is a single-valued map of class $C^1$ \mbox{defined} on a 
neighborhood $U$ of $x_1$, \mbox{$F_1(x_1)=x_2$},
\mbox{$\Lambda_1\in\mathcal{D}(F_1;x_{1},x_{2};S_1)$},
and \mbox{$\Lambda_1=\{DF_1(x_1)\}$}, or
(c)~$\Lambda_1\in\mathcal{D}(F_1;x_{1},x_{2};S_1)$,
$\Lambda_1\cdot T_{x_1}^BS_1\subseteq T_{x_2}^BS_2$ and,
for some coordinate chart $\kappa\!:\!U\!\to\!\rr^{n}$ on $M_2$
such that \mbox{$x_2\!\in\! U$} and $\kappa(x_2)=0$, there is
a \mbox{polyhedral} cone $C$ in $\rr^{n}$ such that 
\mbox{$\kappa(S_2\cap U)=C\cap\kappa(U)$}.}

\vspace{-8pt}
\item[3.] {\bem (The product rule)}  If, for $i=1,2$,  
$F_i\in\mathcal{SVM}(\mathcal{M}^1)$, 
$x_i\!\in\! M_ i\!=\!\so(F_i)$, \mbox{$y_i\!\in\! N_ i\!=\!\ta(F_i)$}, 
$S_i\subseteq M_i$, 
and $\Lambda_i\in\mathcal{D}(F_i;x_{i},y_i;S_i)$, then 
$\Lambda_1\times\Lambda_2$ belongs to 
$\mathcal{D}(F_1\times F_2;(x_1,x_2),(y_1,y_2);S_1\times S_2)$.

\vspace{-8pt}
\item[4.] {\bem ($C^1$ invariance)}  Assume that
(a)~for $i=1,2$,  $M_i,N_i\in\mathcal{M}^1$, 
$F_i\in\mathcal{SVM}(\mathcal{M}^1)$, 
$M_i\!=\!\so(F_i)$, \mbox{$N_i\!=\!\ta(F_i)$}, 
$S_i\subseteq M_i$, $x_i\in M_1$, $y_i\in N_i$, 
$U_i,V_i$ are open
subsets of $M_i,N_i$ such that $x_i\in U_i$ and  $y_i\in V_i$,
(b)~$\Phi:U_1\mapsto U_2$ and $\Psi:V_1\mapsto V_2$ are 
diffeomorphisms of class $C^1$ such that 
$\Phi(x_1)\!=\!x_2$ and $\Psi(y_1)\!=\!y_2$, 
(c)~$\Phi(S_1\!\cap\! U_1)\!=\!S_2\!\cap\! V_2$, and
(d)~the set \mbox{$(\Phi\!\times\!\Psi)\Big(\gr(F_1)\!\cap\!((S_1\cap U_1)
\!\times\!V_1)\Big)$} is equal to
$\gr(F_2)\!\cap\!((S_2\cap U_2)\!\times\! V_2)$. 
Then the set \mbox{$\mathcal{D}(F_2;x_2,y_2;S_2)$} is equal  to
\mbox{$D\Psi(y_1)\circ\mathcal{D}(F_1;x_1,y_1;S_1)
\circ D\Phi^{-1}(x_2)$}. \proofend
\een
\end{definition}

\begin{remark} 
The $C^1$ invariance property implies in particular that a GDT 
$\mathcal{D}$ is {\em local\/}, in the sense that
$\mathcal{D}$ satisfies

\vspace{-8pt}
\ben
\item[4'.] {\bem (Locality)}  Assume that (a)~$M,N\in\mathcal{M}^1$, 
\mbox{$x\in M$}, \mbox{$y\in N$}, $U,V$ is are open
subsets of $M,N$ such that $x\in U$, $y\in V$,
(b)~for $i=1,2$,  \mbox{$F_i\in\mathcal{SVM}(\mathcal{M}^1)$}, 
$M\!=\!\so(F_i)$, \mbox{$N\!=\!\ta(F_i)$}, 
$S_i\subseteq M_i$, \mbox{(c)~$S_1\cap U=S_2\cap U$}
and \mbox{$\gr(F_1)\cap((S_1\cap U)\times V)$}
is equal to \mbox{$\gr(F_2)\cap((S_2\cap U)\times V)$}. Then
$\mathcal{D}(F_1;x,y;S_1)=\mathcal{D}(F_2;x,y;S_2)$.\proofend
\een
\end{remark}

\begin{remark} 
To construct a GDT $\mathcal{D}$, it suffices to \mbox{specify}
$\mathcal{D}(F;0,0;S)$, for all $k,\ell$, 
whenever $F:\rr^k\tto\rr^\ell$
and $S\subseteq\rr^k$. Let $\mathcal{D}_0$ be a correspondence
of the kind described above, but restricted to
$M=\rr^k$, $N=\rr^k$, $x=0$, $y=0$. It is then a routine
matter to write down a list of axioms that will guarantee
that $\mathcal{D}_0$ ``gives rise in a canonical way''
to a GDT in the sense of
Definition \ref{defpfgdt}. 

The crucial point is $C^1$ invariance and locality, i.e., the version
of the $C^1$ invariance condition of Definition \ref{defpfgdt}
restricted to $\rr^k$, $\rr^\ell$. We omit the details of the
general theory, but refer the reader to our discussion of
weak multidifferentials below (in \S \ref{seconwmd}, especially
the remarks following Corollary  \ref{chainrulecor}), where 
we explain how this is done in one particular case. \proofend
\end{remark}

\ssection{The directional open mapping property}\label{domptrans}

\noindent We say that a GDT $\mathcal{D}$ has the {\bem directional 
open mapping property} if the following statement is true:.

\medbreak
\noindent{\bf (DOMP)}\ {\em 
Assume that $n,m\in\zz_+$, 
$F\!\in\!\mathcal{SVM}(\rr^n,\rr^m)$,  $v\!\in\!\rr^m$,
\mbox{$C$ is a closed} convex cone in $\rr^n$, 
$\Lambda$ \mbox{belongs} to $\mathcal{D}(F;0,0;C)$, and 
\mbox{$v\!\in\!\bigcap_{L\in\Lambda}{\rm Int}(LC)$}. Then 
there exists a closed convex cone $D$ in $\rr^n$ such that
$v\in{\rm Int}(D)$, having the property that
for every $\delta\!\in]0,\infty[$ there exists
an $\eps(\delta)\!\in]0,\infty[$ such that
\fbox{$D\!\cap\!\{y\!\in\!\rr^m\!:\!\|y\|\!\le\!\eps(\delta)\}\!\subseteq\!
F(C\!\cap\!\{x\!\in\!\rr^n\!:\!\|x\|\!\le\!\delta\})$}.}\proofend

\ssection{Warga derivate containers (WDCs)}\label{sec-wdc}

\vspace{-2mm}
\begin{definition}\label{tomimic}
Let $f$ be a map from an open subset $\Omega$ of $\rr^m$ to $\rr^n$, and let $x_*\in\Omega$.
A {\em Warga derivate container} of $f$ at $x$ is a nonempty compact 
subset $\Lambda$ of ${\rm Lin}(\rr^m,\rr^n)$ such that 
\ben
\item[(*)] for every open neighborhood $\Lambda'$ of $\Lambda$ 
in the space ${\rm Lin}(\rr^m,\rr^n)$ 
\ there \ exist \ an \ open \ neighborhood
\een 
\skipopag
\ben
\item[]  \mbox{$U=U_{\Lambda'}$ of $x_*$ in $\Omega$} and a sequence
$\{f_j\}_{j=1}^\infty$ of maps of
class $C^1$ from $U$ to  $\rr^n$, such that 
(a)~$f_j\to f$ uniformly on $U$ as $j\to\infty$, and 
(b)~the differential $Df_j(x)$ belongs to $\Lambda'$ for 
every $x\in U$ and every $j$.
\een
\end{definition}

It follows from the definition of a WDC that any map $f$ that admits
a WDC at a point $x_*$ must be Lipschitz continuous on a neighborhood
of $x_*$, since we can \mbox{always} choose $\Lambda'$ to be bounded,
in which case $f$ will be a \mbox{uniform} limit, on a neighborhood
$U$ of $x_*$, of maps of class $C^1$ with uniformly bounded
derivatives.

Associated to the Warga derivate containers we can define  GDTs
by extending the definition of the WDCs to ``WDCs along sets.''
This can be done in more than one way. For example, 
suppose $C$ is a closed convex cone 
in $\rr^n$,  $U$ is a neighborhood of $0$ in $\rr^n$, 
and $f:U\cap C\mapsto\rr^m$ is a Lipschitz continuous
map. We could then define a WDC of $f$ at $0$ in the direction
of $C$ to be a WDC of any extension $\tilde f$ of 
$f$ to a Lipschitz map defined on  a full neighborhood
$V$ of $0$ in $\rr^n$. Alternatively, 
we could mimic Definition \ref{tomimic}, using uniform
approximations of $f$ by maps of class $C^1$
on relative neighborhoods  $V\cap C$ of $0$ in $C$. These
two approaches lead to different theories.

We will not pursue this matter here any further
and will, instead, move on directly
to a more general theory called WMD (``weak multidifferentials'')
that will be defined {\it ab initio} as theory of differentials 
in the direction of sets. 

\vbox{
\ssection{Regular set-valued maps}\label{sec-regsetval}

\noindent If $X$, $Y$ are metric spaces, then 
$SVM_{comp}(X,Y)$ will denote the subset of 
$SVM(X,Y)$ whose \mbox{members} are
the set-valued maps from $X$ to $Y$
that have a \mbox{compact} graph.
We say that a sequence  $\{F_j\}_{j\in\nn}$ of 
\mbox{members} of $SVM_{comp}(X,Y)$ 
{\bem inward graph-converges} to an $F\in SVM_{comp}(X,Y)$---and
write $F_j\togr F$---if
for every open subset $\Omega$ of $X\times Y$ such that 
$\gr(F)\subseteq\Omega$ there exists a
$j_\Omega\in\nn$ such that $\gr(F_j)\subseteq\Omega$
whenever $j\ge j_\Omega$.
}


\begin{definition}\label{defregu} 
Assume that $X$, $Y$ are  metric spaces.  
A {\bem regular set-valued map} from $X$ to $Y$ is a set-valued map 
$F\in{}SVM(X,Y)$ such that 
\begin{itemize}
\vspace{-6pt}
\item{} for every compact subset $K$ of $X$, the restriction
$F\restr K$ of $F$ to $K$ belongs to 
$SVM_{comp}(K,Y)$  and is a limit---in the sense of inward 
graph-convergence---of a sequence of continuous
single-valued maps from $K$ to $Y$. 
\end{itemize}
\vspace{-6pt}\noindent We use $\regu{X}{Y}$ to denote the set of all regular
\mbox{set-valued}  maps from $X$ to $Y$. \proofend
\end{definition}
It is easy to see that if $F:X\mapsto Y$ is an ordinary 
(that is, single-valued and everywhere defined) map, then
$F$ belongs to $\regu{X}{Y}$ if and only if $F$ is continuous.

An important class of examples of regular maps is \mbox{provided} by the
following two results, whose proof we omit.
\begin{theorem}\label{kakucon}
Assume that $K$ is a compact metric 
\mbox{space},  $Y$ is a normed 
space, and  $C$ is a convex subset of $Y$. Let $\Phi:K\tto C$ be a 
set-valued map such that the graph of $\Phi$ is compact and the value 
$\Phi(x)$ is a nonempty convex set for every $x\in K$. Then $\Phi$ is regular
as a map from $K$ to $C$.
\end{theorem}
\begin{theorem}\label{kakucon-nocomp}
Assume that $X$ is a  metric space, $Y$ is a normed 
space, and $C$ is a convex subset of $Y$. Let $\Phi:K\tto C$ be an 
upper semicontinuous set-valued map with nonempty compact
convex values.  Then $\Phi\in\regu{X}{C}$.\proofend
\end{theorem}

In addition, it is not hard to prove the following.
\begin{theorem}\label{comporeg-13}
Assume that $X$, $Y$, $Z$ are metric \mbox{spaces}. Let
\mbox{$F\in\regu{X}{Y}$}, $G\in\regu{Y}{Z}$. Then the
\mbox{composite} map
$G\circ F$ belongs to $\regu{X}{Z}$.\proofend
\end{theorem}

\ssection{Weak multidifferentials}\label{seconwmd}

\noindent ``Multidifferentials'' were studied in \cite{sussmul}. Here we 
\mbox{introduce} a slightly more general theory, of objects that we call
``weak multidifferentials.''

As a preliminary, we need some notations.
If $X$, $Y$ are finite-dimensional real spaces,
$\Omega\subseteq X$, and $\Lambda\subseteq{\rm Lin}(X,Y)$, we use 
$C_{\Lambda}^1(\Omega,Y)$ to denote the set of all maps
$h:\Omega\mapsto Y$ of class $C^1$ such that 
$Dh(x)\in\Lambda$ for all $x\in\Omega$. (The precise meaning
of ``$h$ is of class $C^1$'' is clear if $\Omega$ is open.
We also need to assign a meaning to this expression when
$\Omega$ is closed. This is done in the usual Whitney way,
which turns out to be equivalent to the following: a map
of class $C^1$ from $\Omega$ to $Y$ is a pair $(h_1,H)$ 
consisting of continuous maps \mbox{$h_1:\Omega\mapsto Y$}, 
$H:\Omega\mapsto{\rm Lin}(X,Y)$, having the property
that, for every $x\in\Omega$, 
$h_1(x')-h_1(x)-H(x)\cdot(x'-x)=o(\|x'-x\|)$ as $x'\to x$
via values in $\Omega$. 
If $h=(h_1,H)$ is such an object,
then $Dh$ is defined to be $H$, and if $x\in\Omega$ we define
$h(x)$ to be $h_1(x)$.)

\begin{definition}
Let $X$, $Y$ be finite-dimensional normed real 
\mbox{linear} spaces. Define
$\bar B_X=\{x\in X:\|x\|\le 1\}$,  $Z={\rm Lin}(X,Y)$. Let 
$F:X\tto Y$ be a set-valued map. Let $(x_*,y_*)\in X\times Y$,
and let $\Lambda$ be a compact subset of $Z$. Let $C$ be a 
closed convex cone in
$X$.  We say that $\Lambda$ is a {\sl weak multidifferential of $F$ at
$(x_*,y_*)$ in the direction of $C$}, and write
$$
\Lambda\in WMD(F;x_*,y_*;C)\,,
$$
if the following is true:

\medbreak\noindent(WMD) for every neighborhood $\Lambda'$ of 
$\Lambda$ in $Z$ there exists a pair $(R,\Theta)$ such that
\bi\item[]\bi\bi
\item[(WMD.1)] $R>0$, and $\Theta$ is a function on the interval
$]0,R]$ with values in $[0,\infty[$ and such that 
$\lim_{s\downarrow 0}\Theta(s)=0$, 
\item[(WMD.2)] for every $r\in]0,R]$ there exist  $f,h$  such that
\bi
\item[(WMD.2.1)] $f\in{\rm REG}(x_*+(C\cap r\bar{B}_X),Y)$;
\item[(WMD.2.2)] $\gr(f)\subseteq\gr(F)$;
\item[(WMD.2.3)] $h\in C^1_{\Lambda'}(x_*+(C\cap r\bar{B}_X),Y)$;
\item[(WMD.2.4)]  $h(x_*)=y_*$;
\item[(WMD.2.5)] the inequality  
\ei\ei\ei\ei
\beq\label{issatweak}
\sup\Big\{|\!|y\!-\!h(x)|\!|:y\!\in\!f(x)\Big\}\!\le\!r.\Theta(r)
\eeq 
\bi\item[]\bi\item[]\bi\item[]\bi
holds whenever   $x-x_*\in C\cap r \bar{B}_X$. \mybox
\ei\ei\ei\ei
\end{definition}

Weak multidifferntials satisfy the following  version of the chain rule:

\skipopag

\begin{theorem}\label{chainrule}
Let $X_1,X_2,X_3$ be finite-dimensional \mbox{real} linear spaces, and
let $\bar x_i\in X_i$, $i=1,2,3$. For \mbox{$i=1,2$}, let 
$F_i:X_i\tto X_{i+1}$ be a set-valued map from $X_i$ to $X_{i+1}$, 
let $C_i$ be a closed convex cone in $X_i$, and let
$\Lambda_i$ be %a compact subset of $L(X_i,X_{i+1})$
such that 
$\Lambda_i\in WMD(F_i;\bar x_i,\bar x_{i+1},C_i)$. Let
$S$ be the linear subspace of $X_2$ spanned by $C_2$, and let
$\Pi$ be a linear projection from $X_2$ onto $S$.  Assume that
$F_1(\bar x_1+C_1)$ is a subset of $\bar x_2+C_2$.
Then 
\beq
\Lambda_2\compo\Pi\compo\Lambda_1\in WMD(F_2\compo F_1;\bar x_1,
\bar x_3;C_1)\,.
\eeq
\end{theorem}
Theorem \ref{chainrule} was proved in  \cite{sussmul} for multidifferentials,
but the same proof applies to weak multidifferentials as well and we will
not repeat it here. 

The following is a trivial consequence of Theorem \ref{chainrule}.
\begin{corollary}\label{chainrulecor}
Let $X_1,X_2,X_3$ be finite-dimensional \mbox{real} linear spaces, and
let $\bar x_i\in X_i$, $i=1,2,3$. For \mbox{$i=1,2$}, let 
$F_i:X_i\tto X_{i+1}$ be a set-valued map from $X_i$ to $X_{i+1}$, 
let $C_i$ be a closed convex cone in $X_i$, and let
$\Lambda_i$ be %a compact subset of $L(X_i,X_{i+1})$
such that 
$\Lambda_i\in WMD(F_i;\bar x_i,\bar x_{i+1},C_i)$. Let
$S$ be the linear subspace of $X_2$ spanned by $C_2$, and let
$\Pi$ be a linear projection from $X_2$ onto $S$.  Assume that
$F_1(\bar x_1+C_1)$ is a subset of $\bar x_2+C_2$ and
$LC_1\subseteq S$ for all $L\in\Lambda_1$. Then 
\beq
\Lambda_2\compo\Lambda_1\in WMD(F_2\compo F_1;\bar x_1,
\bar x_3;C_1)\,.
\eeq
\end{corollary}
Using Corollary \ref{chainrulecor}, we can extend WMD to a
GDT by defining $WMD(F;x_*,y_*;S)$, if $M$, $N$ are 
\mbox{manifolds} of class $C^1$, $F:M\tto N$, $x_*\in M$,
$y_*\in N$, and $S\subseteq M$, as follows. We call the set
$S$ {\em conic at $x_*$} if there \mbox{exists} a coordinate chart
$\kappa$ near $x_*$ such that 
$\kappa(x_*)=0$ and $\kappa(\dom(\kappa)\cap S)=
\im(\kappa)\cap C$ for some closed convex cone 
$C$ in $\rr^{{\rm dim}\,M}$.  If $S$ is conic at $x_*$, then
we \mbox{define} $WMD(F;x_*,y_*;S)$  as a set of nonempty compact
subsets of ${\rm Lin}(T_{x_*}M,T_{y_*}N)$  using a chart $\kappa$ as above,
\mbox{together} with a chart for $N$ near $y_*$.
\mbox{Corollary} \ref{chainrulecor} 
then tells us that $WMD(F;x_*,y_*;S)$  does not depend
on the choice of charts, so it is intrinsically defined.
We complete the definition by taking $WMD(F;x_*,y_*;S)$
to be empty if $S$ is not conic at $x_*$. 

It is then easy to show that 
\begin{theorem}
WMD is a GDT that has the DOMP.\proofend
\end{theorem}

\medbreak
\ssection{Generalized differential quotients (GDQs)}\label{sec-gdq}

\begin{definition}\label{defquasi} 
Let $m,n\in\zz_+$, let $F:\rr^m\tto\rr^n$ be a set-valued
map, and let $\Lambda$ be a nonempty compact subset of $\rr^{n\times m}$. 
Let $S$ be a subset of  $\rr^m$.  
We say that $\Lambda$ is a {\bem generalized differential quotient}
(abbreviated ``GDQ'') {\bem of 
$F$ at $(0,0)$ in the direction of $S$}, and write
$\Lambda\in  GDQ(F;0,0;S)$, 
if for every positive real number $\delta$ there exist $U$, $G$ such that 
\ben
\vspace{-6pt}
\item[1.] $U$  is a  compact neighborhood of $0$ in $\rr^m$ and $U\cap S$
is compact;

\vspace{-8pt}
\item[2.] $G$  is a  regular set-valued map from $U\cap S$ to the 
$\delta$-neighborhood  $\Lambda^\delta$ of $\Lambda$ in $\rr^{n\times m}$;

\vspace{-8pt}
\item[3.] $G(x)\cdot x\subseteq F(x)$
for every $x\in U\cap S$.\proofend
\een
\end{definition}

\begin{remark}
The definition of the GDQ is a generalization of a characterization
of classical differentials due to Botsko and Gosser, \cite{botsko}.\proofend
\end{remark}



If $M$, $N$ are $C^1$ manifolds, 
\mbox{$\bar x\!\in\! M$, $\bar y\!\in\! N$, $S\!\subseteq \!M$}, 
and \mbox{$F:M\tto N$}, then we can define a set 
$GDQ(F;\bar x,\bar y;S)$ of nonempty compact linear \mbox{multimaps}
from $T_{\bar x}M$ to $T_{\bar y}N$ 
by picking coordinate charts \mbox{$M\ni x\to\xi(x)\in\rr^m$}, 
\mbox{$N\ni y\to\eta(y)\in\rr^n$}---where \mbox{$m={\rm dim}\,M$,
$n={\rm dim}\,N$}---defined near $\bar x$, $\bar y$ such n 
that $\xi(\bar x)=0$, $\eta(\bar y)=0$, and
declaring a subset $\Lambda$ of ${\rm Lin}(T_{\bar x}M,T_{\bar y}N)$
to \mbox{belong} to $GDQ(F;\bar x,\bar y;S)$  if 
the multimap 
$D\eta(\bar y)\circ\Lambda\circ D\xi(\bar x)^{-1}$ is in $GDQ(\eta\circ
F\circ\xi^{-1};0,0;\xi(S))$.  It turns out that, with this definition,
{\bem the set $GDQ(F;\bar x,\bar y;S)$  does not depend on the choice 
of the charts $\xi$, $\eta$.} Moreover, the following three results
can be proved.

\begin{theorem}\label{isgdt}
GDQ is a generalized differentiation \mbox{theory}
that has the DOMP.\proofend
\end{theorem}

\begin{theorem}\label{contidif}
If $F:\rr^n\mapsto\rr^m$ is a continuous map,
$x\in\rr^n$, and $F$ is classically differentiable at $x$, then
$\{DF(x)\}\in GDQ(F;x,F(x);\rr^n)$.\proofend
\end{theorem}

\begin{theorem}\label{conticlark}
If $F:\rr^n\mapsto\rr^m$ is Lipschitz-continuous,
and $x\in\rr^n$, then the Clarke generalized Jacobian
$\partial F(x)$ belongs to $GDQ(F;x,F(x);\rr^n)$.\proofend
\end{theorem}

\ssection{Comparison of WDC, WMD, and GDQ}\label{sec-compar}

\begin{theorem}\label{contiwmd}
If $F:\rr^n\mapsto\rr^m$ is Lipschitz-continuous,
$x_*\in\rr^n$, and $\Lambda$ is a Warga derivate
container of $F$ at $x_*$, then $\Lambda\in WMD(F\;x_*,F(x_*):\rr^n)$.
\end{theorem}

\noindent {\em Proof.}\ Let $\Lambda'$ be a neighoborhood
of $\Lambda$ in $\rr^{m\times n}$. Let $\{F_j\}_{j=1}^\infty$
be a sequence of $\rr^m$-valued maps of class $C^1$ defined
on a closed ball $B$ in $\rr^n$ with center $x_*$, such that
$F_j\to F$ uniformly on $B$ and $DF_j(x)\in\Lambda'$ for all
$j$ and all $x\in B$. Pick $R$ and $\Theta$ in an arbitrary fashion
so that (WMD.1) holds and $\Theta(s)>0$ when $s>0$. Given $r\in]0,R]$, let 
$f$ be the restriction of $F$ to $B$. Then 
$f$ is a single-valued continuous map, so $f$ is {\it a fortiori}
a regular set-valued map. 

The maps \mbox{$B\ni x\mapsto F_j(x)-F_j(x_*)+F(x_*)=h_j(x)$}
converge uniformly to $f$
on $B$ and satisfy \mbox{$h_j(x_*)=F(x_*)$}. Pick $j$ such that
$\|h_j\|_{sup}\le r.\Theta(r)$, and let $h=h_j$, $C=\rr^n$. 
It is  then clear that all the conditions of (WMD.2)
are satisfied. \proofend

\medbreak Theorems \ref{contidif} and \ref{conticlark}
show that GDQ is a common generalization of both 
the classical theory of differentials and the CGJ. 
Furthermore, it is easy to exhibit maps that have  GDQs at a point $\bar x$
but are not classically 
\mbox{differentiable} at $\bar x$  and do not have 
a CGJ or a WDC. (A simple 
example is provided by the function $f:\rr\to\rr$ 
\mbox{given} by 
$f(x)=x\sin{1/x}$ if $x\neq 0$, $f(0)=0$.  Then
$f$ is not differentiable at $0$ and does not have
a WDC of $f$ at $0$, since $f$ is not Lipschitz continuous 
near $0$. On the other hand, the interval
$[-1,1]$ clearly belongs to $GDQ(f;0,0,\rr)$.)


The previous remarks show that GDQ is ``much more general
than the classical differential and the CGJ combined.''
It would then be natural to conjecture that
GDQ is also more general than WMD, or at least
more general than WDC. This, however, it not true,
because {\em there are Warga derivate containers
that are not GDQs.} 

For a \mbox{simple} example, consider
the function \mbox{$f:\rr\mapsto\rr^2\sim\cc$} \mbox{given} by
$f(x)=\int_0^xe^{\frac{i}{t}}\,dt$. Then $f$ is Lipschitz 
\mbox{continuous}.
Let \mbox{$h_j(t)=t^{-1}$} when 
\mbox{$|t|\ge j^{-1}$}, and $h_j(t)=j^2t$ when \mbox{$|t|\le j^{-1}$}.
Let \mbox{$f_j(x)=\int_o^xe^{i\,h_j(t)}\,dt$}. Then 
\mbox{$f_j\in C^1(\rr,\cc)$},
and $f_j\to f$ uniformly on compact \mbox{sets} as $j\to\infty$.
Furthermore, for 

\skipopag

\noindent every $x$ and \mbox{every} $j$, 
the derivative $f'_j(x)$ belongs to the unit
\mbox{circle} \mbox{$\mathbb{S}^1=\{z\in\cc:|z|=1\}$}.  It follows that 
{\em $\mathbb{S}^1$ is a \mbox{Warga} derivate container of
$f$ at $0$.}  On the other hand, {\em $\mathbb{S}^1$ is not
a GDQ of $f$ at $0$.}  We will actually show that
{\em if $\Lambda$ is any GDQ of $f$ at $0$, then
$0\in\Lambda$}. To see this, assume that
$0\notin\Lambda$, and pick a compact 
\mbox{neighborhood} $\Lambda'$ 
of $\Lambda$ such that $0\notin\Lambda'$. Observe 
that, if $x>0$, then we can make the change of variables
$t=1/s$ and conclude that 
\mbox{$f(x)=\int_{1/x}^\infty\frac{e^{is}}{s^2}\,ds$}.
Integration by parts, with $\xi=1/x$,  then yields
$f(x)=-i\xi^{-2}e^{i\xi}+
2i\int_{\xi}^\infty\frac{e^{is}}{s^3}\,ds$.
A \mbox{second} integration by parts then gives
$$
f(x)=-i\xi^{-2}e^{i\xi}+
2\xi^{-3}e^{i\xi}-6\int_{\xi}^\infty\frac{e^{is}}{s^4}\,ds\,.
$$
But $6|\int_{\xi}^\infty\frac{e^{is}}{s^4}\,ds|\le
6\int_{\xi}^\infty s^{-4}\,ds=2\xi^{-3}$.  Therefore
$|f(x)|=O(\xi^{-2})=O(x^2)$ as $x\downarrow 0$. Now,
since  $\Lambda$ is a GDQ of $f$ at $0$, $f$ must have
a factorization $f(x)=g(x)x$ such that $g$ is continuous
and $\Lambda'$-valued on some punctured neighborhood
$]-\alpha,\alpha[\backslash\{0\}$ of $0$.  Hence there exists
a $\delta\in\rr$ such that $\delta>0$ and
$|g(x)|\ge\delta$ \mbox{whenever} $0<x\le\alpha$. On the other hand,
the bound $|f(x)|=O(x^2)$ implies that
$|g(x)|=O(x)$ as $x\downarrow 0$, and we have reached a 
contradiction.

\ssection{Conclusion}\label{secconclu}
\noindent The remarks at the end of the previous 
section show that the two ``most general'' GDTs 
described so far---that is, GDQ and WMD---do not contain each other. 
In the \mbox{second} paper of this series we introduce a new GDT---the 
``\pigd''---that \mbox{contains} both.


%\bibliographystyle{amsalpha}
%

\skipopag
\end{document}








